Author's response to reviews

Title: Metabolic syndrome among a middle-aged population in the Red River Delta region of Vietnam

Authors:

Binh Tran Quang (binhtq@nihe.org.vn)
Phuong Pham Tran (phuong.pham@nihe.org.vn)
Nhung Bui Thi (nhungvnnin@gmail.com)
Tung Do Dinh (bsdinhtung@gmail.com)

Version: 2 Date: 20 June 2014

Author's response to reviews: see over
June 20, 2014
To: Professor Timothy Shipley
Editor-in-Chief, BMC Endocrine Disorders

Re: Cover Letter

Dear Professor Timothy Shipley,

In the context of relatively little publications on the actual status of metabolic syndrome in the developing countries, the present cross-sectional study is the attempt of the authors to report the prevalence of metabolic syndrome and associated risk factors in a typical rural province in Red River Delta Region of Vietnam.

Please find a manuscript entitled “Metabolic syndrome in a middle-aged population in the Red River Delta region of Vietnam” for the type of research article of the BMC Endocrine Disorders. This manuscript has not been reported in a published paper or is described in a paper submitted or accepted for publication elsewhere before. We also prepared our manuscript based on the instructions of the BMC Endocrine Disorders.

We are looking forward to hearing from you.

Yours sincerely,

Tran Quang Binh, M.D
Department of Immunology and Molecular Biology
National Institute of Hygiene and Epidemiology
1 Yersin, Hanoi, Vietnam
Tel: (84-4) 38214370; Fax: (84-4) 38210853
Email: binhtq@nihe.org.vn